372' Copyright Ron Burback 1997Day-to-Day Cell AdministrationRon BurbackLecture 23

373' Copyright Ron Burback 1997Simple Group dcecp Commandsngroup catalog [-simplename]ngroup delete <group>ngroup rename <group> -to <group>ngroup add <group> -member <principal>ngroup list <group> [-simplename]ngroup remove <group> -member <principal>

374' Copyright Ron Burback 1997Create a Groupngroup create <group> -attribute <list>nwhere the <list> is of the form {{<option> <value>} ...}n<option> <value>n-alias namen-gid numbern-uuid numbern-fullname stringn-inprojlist yes/no

375' Copyright Ron Burback 1997How do I �pause� a group?ngroup modify <group> -change {{-inprojlist no}}nThis will keep the group in the registry with an intact principal list but this group will not be in any PAC.

376' Copyright Ron Burback 1997Can I Extend Group Attributes?ngroup <group> modify -add <ERA>ngroup <group> modify -remove <ERA>nFirst create an ERA and the group then add extended registry attributes.

377' Copyright Ron Burback 1997OrganizationThe organization commands follow the group commands except with a different attribute list.nRecall, the organization defines the policies.nAn organization is a list of principals.nOrganization can have ERAs.

378' Copyright Ron Burback 1997Organization Attributesnorgid numbernuuid numbernfullname sting

379' Copyright Ron Burback 1997Organization Policy Attributesnacctlife [time | unlimited]npwdalpha [yes|no]npwdexpdate [time | none]npwdlife [time | unlimited]npwdminlen numbernpsdspaces [yes|no]

380' Copyright Ron Burback 1997How to force all passwords to be changed (in this org)norganization <org> -modify {{-change pwdexpdate <now>}}

381' Copyright Ron Burback 1997Simple Principal Commandsnprincipal catalog [-simplename]nprincipal delete <principal>nprincipal rename <principal> -to <principal>nprincipal show <principal>

382' Copyright Ron Burback 1997Principal Createnprincipal create <principal> -attribute {attribute value list}nalias [yes|no]nuid numbernuuid numbernfullname stringnquota [number, 0, unlimited]—number of registry objects this principal can create

383' Copyright Ron Burback 1997Simple Accounts Commandsnaccount catalog [-simplename]naccount delete <account>naccount show <account>

384' Copyright Ron Burback 1997Create an Accountnaccount create <principal>n-password <password>n-group <group>n-organization <organization>n-mypassword <password>n-attribute <name value list>

385' Copyright Ron Burback 1997Account Attributesnacctvalid [yes|no]nclient [yes|no]ncreate <date>ndescription <string>ndupkey [yes|no]nexpdate <date>nforwardabletkt [yes|no]

386' Copyright Ron Burback 1997More Account Attributesngoodsince <date>nhome <directory name>nlastchange <principal> <time>npostdatetkt [yes|no]nproxiabletkt [yes|no]npwdvalid [yes|no]nrenewabletkt [yes|no]

387' Copyright Ron Burback 1997More Account Attributesnserver [yes|no]nshell <unix shell>nstdtgtauth [yes|no]nmaxtktlife <time>nmaxtktrenew <time>

388' Copyright Ron Burback 1997Common Account ActionsnFreeze an account?—account <account> -attribute {{acctvalid no}} -mypassword <password>nForce password change?—account <account> -attribute {{pwdvalid no}} -mypassword <password>

389' Copyright Ron Burback 1997Simple Host Commandsnhost catalog [-simplename]nhost ping <hostname>nhost start <hostname>nhost stop <hostname> -forcenhost show <hostname>nhost unconfigure <hostname>

390' Copyright Ron Burback 1997Configure a Hostnhost configure <hostname> -cell <cellname> -client -secmaster <secmastername> -cds <cdsname> -admin <cell_admin> -password <password>

391' Copyright Ron Burback 1997Hostdatanhostdata catalog [-simplename]n/.../cellname/hosts/hostname/config/hostdata/ndce_cf.dbncell_namenpe_sitenhost_namencell_aliasesnpost_processorsnsvr_routing

392' Copyright Ron Burback 1997hostdata createnhostdata create <dataname> -attribute {{uuid <number>} {annotation <string>} {storage <filename>} {hostdata/data <data>}} [-binary]nexample:ndcecp> hostdate create file1 -attribute {{storage /tmp/file1} {data {hello world}}}

393' Copyright Ron Burback 1997hostdata -entrynThe hostdata modify and delete command has the -entry option. If present, the operation works on the name in the cds. If not present, the operation works on the data at the host location.

394' Copyright Ron Burback 1997Common Keytab Commandsnkeytab catalog <keytabname> [-simplename]n/.../cellname/hosts/hostname/keytab/keytabnamenkeytab delete <keytabname> [-entry]

395' Copyright Ron Burback 1997keytab createnkeytab create <keytabname> {-attribute {-uuid <number>} {-annotation <string>} {-storage <filename>} {-data {<list of principals and keys>}}} [-random]n<principal> des [<version> <key>]n<principal> plain <version> <key>

396' Copyright Ron Burback 1997keytab addnkeytab add <keytabname> -member <principal> [ -registry -random -version ]nkeytab add <keytabname> -member <principal> [ -registry ] -key <password> -version <number>

397' Copyright Ron Burback 1997Applicationn<principal, keytab>ncan use the identify of the host machinencan use it�s own identitynobject in the name space—group, profile, rpc object

398' Copyright Ron Burback 1997CDSndirectoriesnlinksnobjects with attributesnclearing houses

399' Copyright Ron Burback 1997Registry nbackupnskulknsink